



**Baishideng  
Publishing  
Group**

7901 Stoneridge Drive, Suite 501,  
Pleasanton, CA 94588, USA  
**Telephone:** +1-925-223-8242  
**Fax:** +1-925-223-8243  
**E-mail:** bpgoffice@wjgnet.com  
<https://www.wjgnet.com>

## **Format for ANSWERING REVIEWERS**

April 21, 2018



Dear Editors,

On behalf of all the authors, I would like to thank you for your consideration of this paper. In the revised manuscript you will find the changes that we made in response to the Reviewers and Editor.

Please find enclosed the edited manuscript in Word format (file name: edited 39011.doc).

**Title: RECENT ADVANCES IN NON-INVASIVE MAGNETIC RESONANCE IMAGING  
ASSESSMENT OF HEPATOCELLULAR CARCINOMA**

**Authors List:** Davide Ippolito· Riccardo Inchingolo, Luigi Grazioli, Silvia Girolama Drago, Michele Nardella, Marco Gatti, Riccardo Faletti

**Name of Journal:** *World Journal of Gastroenterology*

**ESPS Manuscript NO:** 39011

The manuscript has been improved according to Reviewers Comments:

### **Reviewer # 1**

#### **COMMENTS TO AUTHORS**

Q: The overall structure of the manuscript should be improved.

A: We would like to Thank you the reviewer for the consideration of this paper (suggested for High priority for publication) and for your comment. The entire manuscript was carefully read, checked for strengths and weaknesses.

### **Reviewer # 2**

#### **COMMENTS TO AUTHORS**

Q: Good job

A: We would like to Thank you the Reviewer for the positive comment of this paper, and for consideration (acceptance for possible publication).

### **Reviewer # 3**

#### **COMMENTS TO AUTHORS**

Q: ...

A: We apologize for any possible mistake occurred during the preparation of the manuscript, or for any possible uncleared sentence.



**Baishideng  
Publishing  
Group**

7901 Stoneridge Drive, Suite 501,  
Pleasanton, CA 94588, USA  
**Telephone:** +1-925-223-8242  
**Fax:** +1-925-223-8243  
**E-mail:** bpgoffice@wjgnet.com  
**https://** www.wjgnet.com

**Reviewer #4**

**COMMENTS TO AUTHORS**

Q: The article is logical and complete.

A: We would like to Thank you the Reviewer for the positive comment of this paper, and for consideration (acceptance for possible publication).

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*

Sincerely yours,

Davide Ippolito, MD

University of Milano-Bicocca

Department of Diagnostic Radiology, San Gerardo Hospital

Via Pergolesi 33, 20900 - Monza (MB), Italy